Reviewer notes:

Specific Comments to Authors: histopathological details are required why the patient with meningioma would be referred to physiotherapy? did the patient have a hearing problem in the right ear?

1. Histopathological details: Please see last paragraph of case report and appendix A.

2. Patient was referred to speech therapy not physiotherapy. He was referred to the speech therapist as part of rehabilitation and for tracheostomy care to assist with speech, communication and swallowing issues.

3. The patient did not have a hearing issue in his right ear.

4. Figure 1 is an original image.